Author's response to reviews

Title: Sunitinib-induced severe toxicities in Japanese patient with the ABCG2 421 AA genotype

Authors:

Yuji Miura (yujmiura@mac.com)
Chiyo K Imamura (imamuraack@z3.keio.jp)
Koya Fukunaga (koyafukunaga@src.riken.jp)
Yoshiihiko Katsuyama (katuyama@shinshu-u.ac.jp)
Koichi Suyama (kou_susan@yahoo.co.jp)
Toshikazu Okaneya (okaneya@toranomon.gr.jp)
Taisei Mushiroda (mushiroda@riken.jp)
Yuichi Ando (yando@med.nagoya-u.ac.jp)
Toshimi Takano (takano@toranomon.gr.jp)
Yusuke Tanigawara (tanigawara@a7.keio.jp)

Version: 2
Date: 6 September 2014

Author's response to reviews:

Dear Ms. Cherry Battad
Journal Editorial Office
BioMed Central

I have revised our manuscript to include line and page number and acknowledgements section.

Thank you,

Yuji Miura
Department of Medical Oncology
Toranomon Hospital
2-2-2 Minato-ku, Tokyo, Japan
Tel: +81(3) 35881111
Email: yujmiura@mac.com